
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Wed May 15 17:19:40 GMT 2002
Year: 02  Doy: 135
Observer: koon
WEATHER COMMENT: Wed May 15 17:19:42 GMT 2002
Thick cirrostratus overcast, no wind, temp=48F.
**GONG PROBLEM**: Wed May 15 18:01:10 GMT 2002
Going down to the shelter to see why I’m getting the message that the tape
bank is disabled.
**GONG PROBLEM**: Wed May 15 18:44:19 GMT 2002
Back from the shelter.  The problem with the tape drives was baffling, after
consulting with GONG I cycled the power to the Data and Instrument chassis to
clear the problem.  Reason for errors are still unknown.
Wed May 15 18:52:56 GMT 2002    CHIP     Start Patrol
Wed May 15 18:53:05 GMT 2002    PICS     Start Patrol
WEATHER COMMENT: Wed May 15 18:53:09 GMT 2002
Clouds thinned enough fro Spar to track, starting some instruments.
Wed May 15 19:00:56 GMT 2002    CHIP     LSD
Wed May 15 19:03:07 GMT 2002    CHIP     End LSD
Wed May 15 19:03:13 GMT 2002    CHIP     BiasLSD
Wed May 15 19:04:06 GMT 2002    CHIP     End BiasLSD
Wed May 15 19:04:12 GMT 2002    CHIP     Bias
Wed May 15 19:05:03 GMT 2002    CHIP     End Bias
Wed May 15 19:52:10 GMT 2002    CHIP     End Patrol
Wed May 15 19:54:25 GMT 2002    PICS     End Patrol
WEATHER COMMENT: Wed May 15 20:37:51 GMT 2002
Orographic clouds are very thick now, it may rain, closing down.
Wed May 15 20:38:23 GMT 2002 
        MkIV    
 


